UNIVERSITY^{OF} BIRMINGHAM University of Birmingham Research at Birmingham

Comparing antigenaemia- and microfilaraemia as criteria for stopping decisions in lymphatic filariasis elimination programmes in Africa

Stolk, Wilma A.; Coffeng, Luc E.; Bolay, Fatorma; Eneanya, Obiora A.; Fischer, Peter U.; Hollingsworth, T. Deirdre; Koudou, Benjamin G.; Meite, Aboulaye; Michael, Edwin; Prada, Joaquin M.; Rivera, Rocio Caja; Sharma, Swarnali; Touloupou, Panayiota; Weil, Gary J,; de Vlas, Sake J.

DOI:

10.1371/journal.pntd.0010953

License: Creative Commons: Attribution (CC BY)

Document Version Publisher's PDF, also known as Version of record

Citation for published version (Harvard):

Stolk, WA, Coffeng, LE, Bolay, F, Eneánya, OA, Fischer, PU, Hollingsworth, TD, Koudou, BG, Meite, A, Michael, E, Prada, JM, Rivera, RC, Sharma, S, Touloupou, P, Weil, GJ & de Vlas, SJ 2022, 'Comparing antigenaemiaand microfilaraemia as criteria for stopping decisions in lymphatic filariasis elimination programmes in Africa', *PLoS Neglected Tropical Diseases*, vol. 16, no. 12, e0010953. https://doi.org/10.1371/journal.pntd.0010953

Link to publication on Research at Birmingham portal

General rights

Unless a licence is specified above, all rights (including copyright and moral rights) in this document are retained by the authors and/or the copyright holders. The express permission of the copyright holder must be obtained for any use of this material other than for purposes permitted by law.

•Users may freely distribute the URL that is used to identify this publication.

•Users may download and/or print one copy of the publication from the University of Birmingham research portal for the purpose of private study or non-commercial research.

•User may use extracts from the document in line with the concept of 'fair dealing' under the Copyright, Designs and Patents Act 1988 (?) •Users may not further distribute the material nor use it for the purposes of commercial gain.

Where a licence is displayed above, please note the terms and conditions of the licence govern your use of this document.

When citing, please reference the published version.

Take down policy

While the University of Birmingham exercises care and attention in making items available there are rare occasions when an item has been uploaded in error or has been deemed to be commercially or otherwise sensitive.

If you believe that this is the case for this document, please contact UBIRA@lists.bham.ac.uk providing details and we will remove access to the work immediately and investigate.



GOPEN ACCESS

Citation: Stolk WA, Coffeng LE, Bolay FK, Eneanya OA, Fischer PU, Hollingsworth TD, et al. (2022) Comparing antigenaemia- and microfilaraemia as criteria for stopping decisions in lymphatic filariasis elimination programmes in Africa. PLoS Negl Trop Dis 16(12): e0010953. https://doi.org/10.1371/ journal.pntd.0010953

Editor: Marc P. Hubner, University Hospital Bonn: Universitatsklinikum Bonn, GERMANY

Received: March 30, 2022

Accepted: November 14, 2022

Published: December 12, 2022

Copyright: © 2022 Stolk et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: Individual participant data underlying the results presented in this study (Figs 1 and 2 in the main text and in Figs C-E in S1 Supplement) have been posted at the Digital Commons@BeckerLibrary. The URL linking the datasets and metadata is: https://doi.org/10.48765/ nxq0-6562. The model with its source code is made available via gitlab: https://gitlab.com/ erasmusmc-public-health/wormsim.previous. versions. RESEARCH ARTICLE

Comparing antigenaemia- and microfilaraemia as criteria for stopping decisions in lymphatic filariasis elimination programmes in Africa

Wilma A. Stolk^{1*}, Luc E. Coffeng¹, Fatorma K. Bolay^{2†}, Obiora A. Eneanya³, Peter U. Fischer³, T. Déirdre Hollingsworth⁴, Benjamin G. Koudou^{5,6}, Aboulaye Méité⁷, Edwin Michael⁸, Joaquin M. Prada⁹, Rocio M. Caja Rivera⁸, Swarnali Sharma^{10,11}, Panayiota Touloupou^{12,13}, Gary J. Weil³, Sake J. de Vlas¹

 Department of Public Health, Erasmus MC, University Medical Center Rotterdam, Rotterdam, The Netherlands, 2 National Public Health Institute of Liberia (NPHIL), Monrovia, Liberia, 3 Infectious Diseases Division, Department of Medicine, Washington University School of Medicine, St. Louis, Missouri, United States of America, 4 Big Data Institute, Li Ka Shing Centre for Health Information and Discovery, University of Oxford, Oxford, United Kingdom, 5 Centre Suisse de Recherches Scientifiques en Côte d'Ivoire, Abidjan, Abidjan, Côte d'Ivoire, 6 Laboratoire de Cytologie et Biologie Animale, UFR Science de la Nature, Université Nangui Abrogoua Abidjan, Abidjan, Côte d'Ivoire, 7 Programme National de Lutte contre les Maladies Tropicales Négligées à Chimiothérapie Préventive, Abidjan, Côte d'Ivoire, 8 Center for Global Health Infectious Disease Research, University of South Florida, Tampa, Florida, United States of America, 9 Faculty of Health and Medical Sciences, University of Surrey, Guildford, United Kingdom, 10 Department of Biological Sciences, University of Notre Dame, South Bend, Indiana, United States of America, 11 Christian Medical College, IDA Scudder Rd, Vellore, Tamil Nadu, India, 12 Department of Statistics, University of Warwick, Coventry, United Kingdom, 13 School of Mathematics, University of Birmingham, Birmingham, United Kingdom

† Deceased.

* w.stolk@erasmusmc.nl

Abstract

Background

Mass drug administration (MDA) is the main strategy towards lymphatic filariasis (LF) elimination. Progress is monitored by assessing microfilaraemia (Mf) or circulating filarial antigenaemia (CFA) prevalence, the latter being more practical for field surveys. The current criterion for stopping MDA requires <2% CFA prevalence in 6- to 7-year olds, but this criterion is not evidence-based. We used mathematical modelling to investigate the validity of different thresholds regarding testing method and age group for African MDA programmes using ivermectin plus albendazole.

Methodolgy/Principal findings

We verified that our model captures observed patterns in Mf and CFA prevalence during annual MDA, assuming that CFA tests are positive if at least one adult worm is present. We then assessed how well elimination can be predicted from CFA prevalence in 6-7-year-old children or from Mf or CFA prevalence in the 5+ or 15+ population, and determined safe (>95% positive predictive value) thresholds for stopping MDA. The model captured trends in Mf and CFA prevalences reasonably well. Elimination cannot be predicted with sufficient

Funding: The authors WAS, LEC, TDH, EM, JMP, RMCR, SS, PT and SJdV gratefully acknowledge funding of the NTD Modelling Consortium by the Bill & Melinda Gates Foundation [OPP1184344, https://www.gatesfoundation.org]. WAS, FKB, OAE, PUF, BGK, AM, GJW and SJdV, acknowledge funding from Bill & Melinda Gates Foundation [OPPGH5342, https://www.gatesfoundation.org]. LEC further acknowledges funding from the Dutch Research Council NOW [grant 016.Veni.178.023, https://www.nwo.nl/]. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: I have read the journal's policy and the authors of this manuscript have the following competing interests: The filarial antigen test used in this study uses reagents licensed from Barnes-Jewish Hospital, an affiliation of Gary. J. Weil. All royalties from sales of these tests go to the Foundation for Barnes-Jewish Hospital (https://www.foundationbarnesjewish.org/), a registered not-for-profit organization.

certainty from CFA prevalence in 6-7-year olds. Resurgence may still occur if all children are antigen-negative, irrespective of the number tested. Mf-based criteria also show unfavourable results (PPV <95% or unpractically low threshold). CFA prevalences in the 5+ or 15+ population are the best predictors, and post-MDA threshold values for stopping MDA can be as high as 10% for 15+. These thresholds are robust for various alternative assumptions regarding baseline endemicity, biological parameters and sampling strategies.

Conclusions/Significance

For African areas with moderate to high pre-treatment Mf prevalence that have had 6 or more rounds of annual ivermectin/albendazole MDA with adequate coverage, we recommend to adopt a CFA threshold prevalence of 10% in adults (15+) for stopping MDA. This could be combined with Mf testing of CFA positives to ensure absence of a significant Mf reservoir for transmission.

Author summary

Mass drug administration (MDA) of antifilarial drugs is the main strategy towards the elimination of lymphatic filariasis (LF), but defining when MDA can safely be stopped is challenging. Current stopping guidelines require that the prevalence of microfilaraemia (Mf) in the population falls below 1% or the prevalence of circulating filarial antigenaemia (CFA) in 6-and-7-year-old children falls below 2%. The evidence base underlying this threshold is limited. The accuracy of this threshold is hard to assess in field studies due to the long timespan between stopping MDA and the occurrence of either elimination or resurgence. We used mathematical modelling to assess how well elimination can be predicted at community-level from the Mf or CFA prevalence observed 1-year after the last MDA round and we determined safe stopping thresholds (>95% certainty that elimination is achieved). We found that the currently used CFA prevalence in 6–7 year-old children is a poor indicator for stopping decisions in LF elimination programmes implementing MDA with ivermectin plus albendazole, and that CFA prevalence among people aged 5 or 15 years and older is a much more reliable and practical indicator of elimination outcomes. Lymphatic filariasis elimination programmes should reconsider guidelines for stopping MDA.

Introduction

Lymphatic filariasis (LF) is a mosquito-transmitted parasitic worm infection that prevails in tropical and subtropical regions across the world and is an important preventable cause of morbidity and disability [1]. Most LF is caused by the filarial nematode *Wuchereria bancrofti*. The long-lived adult worms reside in the human lymph system and are the main cause of morbidity (lymphoedema and hydrocele). Fertilized female worms release large numbers of microfilariae (worm's offspring) into the bloodstream, where they can be picked up by mosquitoes to be passed on to other humans. Annual mass drug administration (MDA) is the mainstay of control, sometimes complemented by vector control measures such as the use of insecticide-treated bed nets.

The Global Programme to Eliminate Lymphatic Filariasis (GPELF) was initiated in 2000. It aims to eliminate LF as a public health problem by interrupting transmission through MDA of

anti-filarial drugs (albendazole with ivermectin and/or diethylcarbamazine, with the threedrug combination restricted to specific settings warranting acceleration[2]) and by alleviating the suffering of people with clinical manifestations through morbidity management and disability prevention [3,4]. The rapid scale-up of MDA has led to a massive decline in the prevalence of LF infection. By 2020, 17 previously endemic countries no longer require MDA after reaching prevalence targets below which transmission is thought to be unsustainable [1]. Eight more have stopped all MDA and are now under surveillance to confirm elimination as a public health problem based on WHO criteria; 47 countries still require MDA in some or all of their endemic areas [5].

Progress towards elimination of W. bancrofti is monitored by examining trends in microfilaraemia (Mf) or circulating filarial antigenaemia (CFA) prevalence [6]. Based on experiences from China [7], a Mf prevalence below 1% is thought to lead to elimination where infection is transmitted by Anopheles or Culex mosquito species [6]. A lower threshold target is recommended where more efficient *Aedes* species transmit the parasites [6]. However, Mf prevalence surveys require microscopic examination of blood samples, and are extra cumbersome in areas such as Africa where night blood samples are needed because of nocturnal Mf periodicity. In contrast, CFA can be detected in day or night blood with a rapid point of care test that does not require microscopy [8,9]. The World Health Organization (WHO) therefore defined its stop-MDA recommendations based on CFA prevalence, which is to be assessed in 6-7-year-old children according to a specifically designed transmission assessment survey (TAS) [6]. MDA can be stopped if CFA prevalence is <2% where Anopheles or Culex are the main vectors and <1% where *Aedes* is the main vector [6]. Children are taken as sentinel group, as they should be free of LF infection if transmission is successfully interrupted. TAS is to be repeated two more times with intervals of two to three years between surveys, to validate elimination as a public health problem.

The TAS design was found to be practical, but whether this is sufficient to eliminate transmission and sustain the results in the long run remains uncertain [10]. There is concern that CFA prevalence in children is not a sensitive enough indicator for detecting residual transmission [11]. Children are not always exposed to the same number of infective mosquitoes as adults and may be CFA negative while transmission is ongoing in the adult population. Also, there is concern that the currently-used threshold is too stringent when the Alere Filariasis Test Strip (FTS) is used to detect CFA, because it is more sensitive than the previously used immunochromatographic card test (ICT); many FTS-positives are Mf negative and are therefore unlikely to contribute to transmission [12,13].

Mathematical models are useful tools to theoretically assess the validity of MDA stopping thresholds [14,15]. Modelling showed that the risk of LF resurgence is low in African settings where Mf prevalence in the 5+ population was reduced to <1% after five rounds of MDA with ivermectin plus albendazole [15]. Here, we use the established LYMFASIM model to assess the predictive value of Mf and CFA prevalence thresholds for the eventual occurrence of resurgence or elimination of LF, in relation to the sampled age group, for African LF elimination programmes implementing annual MDA with ivermectin plus albendazole. Before doing so, we verify that the model adequately captures the observed association between community-level Mf and CFA prevalence, before and during annual MDA.

Methods

The LYMFASIM model

LYMFASIM is an individual-based model, simulating LF transmission in a population (village or town) of hypothetical individuals [16,17]. The model tracks infection intensity (number of

male and female adult worms; Mf density) for each individual, accounting for inter-individual variation in exposure to infection and participation in interventions. It captures the key processes involved in parasite biology, transmission and interventions. See section 1 in S1 Supplement for more information. LYMFASIM accounts for measurement variation in counting Mf, allowing for the occurrence of false-negative Mf counts for individuals with low worm burdens. For the current analysis, we assumed that all individuals with \geq 1 mature male or female worm are CFA-positive, and that people turn CFA-negative within one month after a person's last worm has died.

Validation of model-predicted CFA prevalences

We validated model-predicted CFA prevalences using individual-level data from annuallytreated communities from community intervention trials in Côte d'Ivoire (primary results will be published elsewhere) and Liberia[18] evaluating the effectiveness of annual versus biannual MDA on LF infection. Only for validating the model-predicted association between Mf and CFA prevalences at baseline (2013 for Côte d'Ivoire, 2014 for Liberia) we also used baseline data from biannually-treated communities.

We simulated the local history of control at community level, accounting for the timing of MDA rounds and surveys, and for the local bed net coverage since 2006. To obtain simulation runs across the spectrum of observed baseline endemicity levels, we performed a large number of runs per scenario, varying model parameters relating to setting-specific transmission conditions: the monthly biting rate, the degree of interindividual variation in exposure to mosquito bites, and external force-of-infection (i.e. the rate at which infection are acquired from outside the simulated population due to human or vector mobility). The external force-of-infection was set to zero in half of the simulation runs, to mimic communities where transmission is independent of imported infections. In the other runs, it was set to a low value, varying between runs but constant over time to mimic communities where low endemicity is stabilized by incoming infection from surrounding areas.

We then checked whether the model-predicted association between mf and CFA prevalences matched to village-level data at baseline and after the first, second and third annual treatment. We further assessed whether the model adequately captures time trends in infection. Lastly, we compared model-predicted and observed age-patterns in infection for all measurements moment, aggregating data from different communities by country. A more detailed description of the data and validation methods is provided in section 2 in <u>S1 Supplement</u>.

Predictive values of Mf and CFA prevalence for elimination and resurgence

We assessed how well the eventual occurrence of elimination of transmission can be predicted based on the Mf or CFA prevalence one year after the last treatment round. We simulated four MDA scenarios with either six or eight annual rounds of MDA with ivermectin plus albendazole, and either 60% or 80% treatment coverage of the total population (with treatment restricted to individuals aged 5 years or older). Bed net use was not considered in these simulations, as we were interested in the dynamics of resurgence and elimination in the absence of bed nets.

This analysis was done for settings with an average population size of about 1000 individuals and a moderate to high baseline Mf prevalence (i.e. between 20% and 40%). We only considered settings without importation of infection from surrounding areas (external force of infection = 0), so that elimination outcomes were not influenced by the presence or absence of importation from surrounding areas. Between runs we varied the monthly biting rate and the shape of the gamma distribution describing variation in exposure between individuals. Parameters were sampled from the parameter space presented in Fig A in <u>S1 Supplement</u> (left panel) and were accepted when the resulting pre-control Mf prevalence fell into one of the 1%-width bins between 20% and 40%; the model was run until we had 100 parameter combinations per bin, resulting in 2000 runs per scenario.

For each run, we recorded predicted time trends in Mf and CFA prevalence in the population aged 5 or 15 years and above, as well as the CFA prevalence among 6-7-year-old children, from a year before the first treatment until 50 years after the last treatment round. Elimination was defined as zero Mf prevalence in the 5+ population 50 years after the last treatment round. To assess to which extent the eventual occurrence of elimination can be predicted based on Mf and CFA prevalence surveys done 1 year after the last treatment, we calculate receiver operating characteristic (ROC) curves for a range of prevalence thresholds, assuming that all individuals per age group are examined. For the calculation of PPV and NPV values, we assumed that a random binomial sample of 200 individuals in the respective age group was surveyed (sampled with replacement); as the number of 6-7-year olds is much lower than this, this implies that nearly all individuals in this age group are included in the survey. This sampling was repeated 100 times to smooth the resulting predictions. Stop-MDA thresholds were estimated at a PPV of 95%. A sample size of 200 was chosen, assuming that this would be a practically feasible number, but alternative values were also used in a sensitivity analysis.

Lastly, we assessed how assumptions regarding the rate of Mf production per worm (which influences the association between Mf density and the underlying number of worms) affect the predicted elimination thresholds. The Mf productivity rate was doubled or halved from 0.58 (estimated previously by fitting model to data)[17] to 1.16 or 0.29 Mf/ μ L blood per female worm per month); to counter the impact on transmission intensity, we simultaneously halved or doubled the mosquito biting rates (see Fig B in S1 Supplement).

PRIME-NTD principles

In <u>S1 Supplement</u> section 3, we describe our adherence to the five principles of the NTD Modelling Consortium on good practice for policy-relevant modelling [19].

Results

Comparison of model-predicted and observed Mf-CFA prevalence associations

Fig 1 shows that model-predicted Mf-CFA prevalence combinations before the introduction of MDA matched well to observed data. The FTS-based observed CFA prevalences from Côte d'Ivoire were relatively high, considering the corresponding Mf prevalences, but could be matched by the model by including importation of infection from surrounding areas. The ICT-based observed CFA prevalences for Liberia were often relatively low for given Mf prevalences and could often be captured without assuming infection importation.

Fig 2 shows that the model also adequately captured the association between communitylevel Mf and CFA prevalence in Côte d'Ivoire during MDA (11 months after the first, second and third treatment round). However, the predicted Mf and CFA prevalence levels per village for the follow-up moment did not always match the observations. Deviations can result from sampling variation in the observed data (sometimes leading to erratic patterns in the data that are not mimicked by the model) or from a mismatch between assumed and actual coverage of MDA (true coverage is unknown and can vary over time and between communities). Fig C in <u>S1 Supplement</u> presents similar results for Liberia, which are somewhat more difficult to interpret due to a switch in CFA diagnostic from ICT at baseline to FTS in later surveys, with the





https://doi.org/10.1371/journal.pntd.0010953.g001

latter resulting in somewhat higher CFA prevalences (indicated by observations from communities were both tests were used in parallel). At baseline, the observed prevalences from Liberia were usually located within the band of predicted values, whereas they were close to the upper edge of the model predictions band or even above it at the three follow-up moments. This was most extreme after the third treatment round, when Mf-prevalence was zero in all communities but CFA prevalence up to 13% were observed. Various factors could contribute to the deviation between model predictions and observations, including uncertainty about MDA coverage and treatment effects on adult worms. A lag in antigen clearance after worm killing



Fig 2. Observed and simulated prevalence of microfilaraemia (Mf) and circulating filarial antigenaemia (CFA, measured by filarial test strip) for Côte d'Ivoire, at baseline (2014, before the first treatment round) and in 2015, 2016, and 2017 (i.e. 11 months after the first, second and third annual MDA rounds). Observations are shown as coloured markers with 95% confidence intervals. Model predictions are shown as small dots. Simulation results from runs matched to specific villages at baseline are shown in the colour of that village, and all other runs are shown in grey. A run was considered a match if the predicted Mf-CFA prevalence combination at baseline fell within the ellipse drawn around the observed MF-CFA prevalence estimates are presented in the figures (i.e. not age-standardized). The MDA coverage was assumed to be 65% of the total population per round in the simulation runs. See Table B in S1 Supplement for details about the simulated scenario and see Fig C in S1 Supplement for a similar figure for Liberia.

https://doi.org/10.1371/journal.pntd.0010953.g002

could theoretically also explain some of these results. However, due to the long time-interval between treatment and the follow-up surveys, this is not the most likely explanation. Also, if this plays a role, we would expect it to be visible in both countries and not only in Liberia. The observed reductions in Mf and CFA prevalence were larger than predicted by the model, for most villages, again suggesting a mismatch between assumed and actual coverage of MDA. Still, for both Liberia and Côte d'Ivoire the observed Mf and CFA prevalence combinations were all on or near the model-predicted associations.

Figs D and E in <u>S1 Supplement</u> show that the model also captured the age-patterns in Mf and CFA prevalence, including the considerably lower prevalence levels in children compared to adults.

Predictive performance of MF- and CFA-based stop-MDA thresholds

Model-predicted trends in Mf and CFA prevalence during and after MDA (six or eight rounds, 60% of 80% coverage) are shown in Figs F and G in S1 Supplement for settings with baseline Mf prevalence between 20%-30% and 30%-40% in the population aged 5 years and above. Longer duration of MDA and higher coverage resulted in lower Mf and CFA prevalence one year after the last treatment and higher probability of elimination. The ROC curves in Fig 3 show how well the eventual occurrence of elimination could be predicted from Mf and CFA prevalence one year post-MDA, for different age groups (assuming that all individuals per age group are included in the survey) and for a range of different threshold values. Adopting a higher threshold value results in a higher sensitivity (i.e. a higher proportion of runs ending in elimination correctly classified as such, based on a prevalence below the chosen threshold), but also a lower specificity (i.e. a lower proportion of runs ending in resurgence correctly classified as such, based on a prevalence above the chosen threshold). This lower specificity means a higher probability of falsely declaring elimination and prematurely stopping MDA, the most important adverse outcome. In the scenario with six treatment rounds with 60% coverage, the Mf prevalence in the 5+ or 15+ population seemed to be a slightly better indicator of elimination than CFA prevalence in any of the age groups considered. However, the overall probability of elimination was still low in this scenario (43% of all runs). The predictive value of CFA prevalence in the 5+ or 15+ population increased with longer MDA and higher coverage, and hence higher elimination probability. The CFA prevalence in the 5+ and 15+ population clearly were the best predictors in the scenario with eight treatment rounds and 80% coverage and the CFA prevalence among 6-7-year olds was the worst predictor, independently of the threshold considered. In this scenario, about 87% of runs resulted in elimination. These patterns remained when the results were further stratified by endemicity, based on baseline Mf prevalence in the 5+ population (Fig H in S1 Supplement).

Fig 4 shows per indicator how the PPV (probability of achieving elimination within 50 years after the last round of MDA if the 1-year post MDA prevalence in a sample of 200 individuals per age group was below a given threshold) depended on the chosen threshold. The PPV never reached 100% for CFA prevalence among 6-7-year olds and Mf prevalence in samples of the 5+ or 15+ population. This means that there was still a risk of resurgence, even when the simulated CFA or Mf prevalence was zero. A PPV value of 100% could only be achieved with the stop criterion based on CFA prevalence in the 5+ or 15+ population. The threshold prevalence associated with a 95% PPV was about 0.5% for the Mf prevalence in the 15+ population, if reached at all. The thresholds based on CFA prevalence in the 5+ and 15 + population were much higher, i.e. 8% and 10% respectively. The corresponding NPV (probability of recrudescence if prevalence was above the threshold) declined from nearly 60% in the scenario with six MDA rounds with 60% coverage to about 25% in the scenario with eight MDA rounds with 30–40% baseline Mf prevalence than in runs with 20–30% baseline Mf prevalence (Fig J in S1 Supplement).

The PPV for a given Mf or CFA prevalence threshold increases with the number of people sampled and thresholds associated with 95% PPV shift to higher values (Fig K in <u>S1 Supplement</u>). Mf prevalence thresholds still remain low, and the PPV for CFA prevalence among 6-



Fig 3. Receiver-operator characteristic (ROC) curves for predicting the eventual occurrence of elimination of transmission under different MDA scenarios, based on the simulated microfilaraemia (Mf) or circulating filarial antigenaemia (CFA) prevalence 1 year after the last treatment round. Different lines show the predictive performance of the Mf and CFA prevalence assessed in all people aged 5 or 15 years and above (Mf: red and grey; CFA: green and pink) and the CFA prevalence in all 6-7-year-old children (blue). Sensitivity is the percentage of simulation runs ending in elimination that are correctly identified based on Mf or CFA prevalence below a range of thresholds (see legend). Similarly, 100%-specificity is the percentage of simulation runs that is falsely classified as having achieved elimination, out of all runs that did not result in elimination, resulting in premature stopping of MDA. The optimal situation is in the upper left corner of the panels (100% sensitivity and 100% specificity). Results are based on 2000 simulations per scenario, with baseline Mf prevalence varying between 20% and 40%.

https://doi.org/10.1371/journal.pntd.0010953.g003

7-year olds never exceeds 90%. For a given sample size, measuring CFA or Mf in the 15+ population results in slightly higher PPV and higher threshold values than measuring CFA prevalence in the 5+ population, because the prevalence of infection is lower in 5–15 year old children. Halving or doubling Mf-productivity (while doubling or halving the biting rate) had hardly any impact on the pre-control Mf-CFA prevalence association and the positive predictive value of the suggested thresholds (Figs L and M in S1 Supplement).



Threshold for prevalence (%)

Fig 4. Probability of achieving elimination within 50 years after the last round of MDA, if the 1-year post MDA prevalence was below a given threshold (i.e. the positive predictive value of using this threshold). Different lines show the predictive performance of the Mf and CFA prevalence in a sample of 200 individuals taken from the 5+ or 15+ population (MF: red and grey; CFA: green and pink) and the CFA prevalence in a sample of 200 6-7-year-old children (blue). Results are based on 2000 simulations per scenario, with baseline Mf prevalence varying between 20% and 40%. The PPV curves based on CFA prevalence in the 5+ or 15+ population are somewhat erratic at low prevalences in the panel for settings that had 6 rounds of MDA with 60% coverage. This is explained by the fact that only few simulation runs ended with such low CFA prevalences in this treatment scenario (i.e. the denominator for calculating the PPV). If one of these few runs ended in resurgence, it causes relatively large fluctuations in the estimated PPV.

https://doi.org/10.1371/journal.pntd.0010953.g004

Discussion

We showed that the eventual occurrence of elimination cannot be predicted with sufficient certainty from the CFA prevalence measured among 6-7-year-old children, irrespective of the number tested, nor from the Mf prevalence measured in a sample of up to 200 individuals from the 5+ or 15+ population. While the predictive value of an Mf-based stop-MDA threshold would improve if based on a full-population survey, thresholds remain low, requiring high (and more costly) sampling sizes. Elimination can be predicted with much more certainty

based on CFA prevalence in the 5+ or 15+ population, which has the added advantage of considerably higher thresholds and thereby smaller sample sizes needed. Focusing on the 15+ age group may be preferred, as a high PPV can be achieved with higher thresholds and thus lower sample sizes than when focussing on the entire 5+ population. It has the added advantage that 5–14 year old children don't need to be tested, which could make the surveys more acceptable to the community and easier to execute. A CFA prevalence of 10% in this population subgroup corresponds to a PPV of 95% or higher, and this threshold is robust for endemicity level and number of MDA rounds.

The proposed CFA threshold, chosen for its high PPV (\geq 95%), is associated with a relatively low NPV (i.e. the probability of resurgence if the prevalence is above the threshold), with NPV values between 25% and 60% depending on the treatment scenario. We consider this acceptable as the consequence of a false negative outcome (continuing MDA for too long at the cost of wasting drugs and other resources) is less of a problem than a false positive result (stopping MDA too early and having to reinstate the MDA infrastructure).

To reproduce the pre- and post-MDA Mf-CFA prevalence association with our model, it was sufficient to assume that CFA positivity occurs when an individual has at least one mature worm, even though village-specific time trends in Mf and CFA prevalence were not always matched exactly. Some unexplained variation remains in the data, which may be due to sampling variation or other factors not captured in the model. For example, direct comparison of the ICT card test and the Filarial Test Strip (FTS) for measuring antigenaemia suggested that the latter is more sensitive, especially in low endemic settings [13]. Accounting for this would bring the model closer to the relatively low ICT-based CFA prevalences observed in Liberia (Fig 1). Also, accounting for the impact of ivermectin MDA for onchocerciasis prior to the trial onset in Côte d'Ivoire, could help to reproduce observations with relatively low Mf prevalence and high CFA prevalence, as ivermectin causes a strong reduction in Mf density but does not kill adult worms [20,21]. Further refinement of the diagnostic model, e.g. to capture temporal dynamics of antigenaemia density after worm-death or a possible differential contribution of male and female worms to antigenaemia density, requires more-detailed individuallevel infection and treatment data. More detailed modelling of antigenaemia decay rates would likely have little impact on model predicted antigenaemia prevalences 12 months after treatment, on which our threshold estimates were based. However, even If CFA tests remain positive that long after treatment-induced death of the last worm, this would not change the main findings, but would only result in slightly higher threshold values.

Uncertainty is inherent to modelling, e.g. concerning the values of model parameters or model structure. We used the LYMFASIM model, which was quantified for *Anopheles*-transmitted bancroftian filariasis in Africa and was validated in this study against data from West Africa. It is reassuring that our PPV and NPV estimates for CFA prevalence among 6-7-year olds are fairly similar to those estimated with another simulation model [22]. Their estimates for Malindi (a Kenyan village with baseline endemicity within our simulated range) are comparable to our estimates from the most intensive MDA scenario (with eight MDA rounds and 80% coverage), both with respect to the overall probability of elimination and the PPV of different infection indicators. We anticipated that our predictions would be influenced by assumptions about the ratio of Mf to adult worm density. We explored the impact of using a different Mf productivity rate, but found that doubling or halving this value (and thus halving and doubling corresponding worm burdens) did not change the overall conclusions regarding the relative performance of different indicators for stopping. Still, in-depth analysis of individual-level Mf and CFA positivity or density data could help to quantify such uncertain parameters to further improve the accuracy of our model [23].

WHO chose to assess infection in 6-7-year-old children, as any infection in this age group must be recently acquired, meaning that transmission has not been completely interrupted [6]. Focussing on this age group is also practical, as it allows for school-based sampling. The 2% CFA prevalence threshold was assumed to be "a conservative proxy for an Mf prevalence of <1%", considering the higher sensitivity of CFA tests for detecting infection [6]. However, our study clearly demonstrates that CFA prevalence in this age group is not the most informative predictor of the eventual achievement of elimination. As 6-7-year olds constitute a small proportion of the population and are less exposed than older individuals, even a zero prevalence in this group does not preclude eventual resurgence of LF in the population (demonstrated by the model-estimated PPV values in this age group never exceeding 90%).

Our simulations were specific for African areas where LF is transmitted by *Anopheles*-species and ivermectin plus albendazole is the MDA treatment regimen. Our thresholds are not valid for areas, where LF is transmitted by different vector species (e.g. *Culex* or *Aedes*). Especially the *Aedes*-vectors are known to be highly efficient, which led WHO to propose lower stop-MDA thresholds for areas with *Aedes* transmission. The employed treatment regimen can also influence the post-MDA CFA-Mf association and hence the CFA-thresholds. However, our conclusions about the insufficient sensitivity of TAS in 6-7-year-old children are generalizable. Indeed, concerns about the reliance on measures in such young children were voiced before, when evidence of persistent LF transmission was found after passing TAS and stopping MDA, e.g. in Sri Lanka and American Samoa [24–27]. Detailed studies in these countries carried out several years after passing TAS and stopping MDA, confirmed that CFA prevalence in such young children was considerably lower than the prevalence among older individuals or adults only [11,28]. Moreover, these studies demonstrated that community-based TAS is feasible, although it is logistically more difficult, takes more time and is more expensive.

Our quantitative CFA prevalence threshold estimates (10% in 15+) should be interpreted with caution, also for the African context. They are based on modelling and our long-term elimination predictions remain to be validated. These quantitative thresholds apply to communities with moderate to high baseline Mf prevalence (between 20% and 40%) and under the specified transmission and treatment conditions. In reality there may be more variation in transmission conditions, treatment history, or age-infection patterns, which can influence the safe threshold values. For example, lower thresholds may be needed in settings where low pre-MDA Mf prevalence is stabilized by continued imported infection from surrounding areas or strong aggregation of exposure, as was shown in other theoretical studies for filarial infections [22,29,30]. To obtain more certainty about the stop-MDA decision, presence of Mf can be assessed in CFA positives to ensure absence of a significant Mf reservoir for transmission. This is feasible, as night blood sampling and processing would be required for a limited number of people. If stop-MDA decisions are made for larger geographical areas, encompassing multiple communities with varying baseline endemicities, thresholds should be adjusted to account for heterogeneity within the area. This is most obvious when the area includes non-endemic communities, which cause dilution of infection levels for the endemic part of the area, requiring lower threshold levels.

We conclude that the currently used CFA prevalence in 6–7 year-old children is a poor indicator for stopping decisions in LF control by MDA if elimination of transmission is the eventual goal. We recommend using CFA prevalence of the 15+ population at a threshold of 10% (with Mf follow-up) as critical threshold for stopping MDA in LF elimination programmes where treatment has been provided for at least 5 or 6 years with adequate coverage, for application at the level of communities or somewhat larger areas with relatively homogeneous transmission conditions.

Supporting information

S1 Supplement. PDF file with supplementary information to the manuscript. The document has the following sections: section 1) LYMFASIM: model and parameter values; section 2) Validation of model-predicted CFA prevalence levels; section 3) PRIME-NTD table: Policy-Relevant Items for Reporting Models in Epidemiology of Neglected Tropical Diseases; section 4) Detailed results. (PDF)

Author Contributions

- **Conceptualization:** Wilma A. Stolk, Luc E. Coffeng, T. Déirdre Hollingsworth, Sake J. de Vlas.
- Data curation: Fatorma K. Bolay, Obiora A. Eneanya, Peter U. Fischer, Benjamin G. Koudou, Aboulaye Méité, Gary J. Weil.
- Formal analysis: Wilma A. Stolk, Luc E. Coffeng, Sake J. de Vlas.
- Funding acquisition: Peter U. Fischer, T. Déirdre Hollingsworth, Gary J. Weil.
- Investigation: Wilma A. Stolk, Luc E. Coffeng, T. Déirdre Hollingsworth, Joaquin M. Prada, Rocio M. Caja Rivera, Swarnali Sharma, Panayiota Touloupou, Gary J. Weil, Sake J. de Vlas.
- **Methodology:** Wilma A. Stolk, Luc E. Coffeng, T. Déirdre Hollingsworth, Edwin Michael, Joaquin M. Prada, Rocio M. Caja Rivera, Swarnali Sharma, Panayiota Touloupou, Sake J. de Vlas.
- Project administration: Sake J. de Vlas.
- Software: Wilma A. Stolk, Luc E. Coffeng.
- Supervision: Peter U. Fischer, Gary J. Weil, Sake J. de Vlas.
- Validation: Luc E. Coffeng.
- Visualization: Wilma A. Stolk, Luc E. Coffeng, Sake J. de Vlas.
- Writing original draft: Wilma A. Stolk.
- Writing review & editing: Wilma A. Stolk, Luc E. Coffeng, Obiora A. Eneanya, Peter U. Fischer, T. Déirdre Hollingsworth, Benjamin G. Koudou, Aboulaye Méité, Edwin Michael, Joaquin M. Prada, Rocio M. Caja Rivera, Swarnali Sharma, Panayiota Touloupou, Gary J. Weil, Sake J. de Vlas.

References

- (2020) Global programme to eliminate lymphatic filariasis: progress report, 2019. Wkly Epidemiol Rec 95: 509–524.
- World Health Organization (2017) Guideline: alternative mass drug administration regimens to eliminate lymphatic filariasis. Geneva: World Health Organization. WHO/HTM/NTD/PCT/2017.07 WHO/HTM/ NTD/PCT/2017.07.
- Ottesen EA (2000) The global programme to eliminate lymphatic filariasis. Trop Med Int Health 5: 591– 594. https://doi.org/10.1046/j.1365-3156.2000.00620.x PMID: 11044272
- Addiss DG, Brady MA (2007) Morbidity management in the Global Programme to Eliminate Lymphatic Filariasis: A review of the scientific literature. Filaria J 6: 2. https://doi.org/10.1186/1475-2883-6-2 PMID: 17302976

- (2021) Global programme to eliminate lymphatic filariasis: progress report, 2020. Wkly Epidemiol Rec 96: 497–508.
- World Health Organization (2011) Monitoring and epidemiological assessment of mass drug administration in the global programme to eliminate lymphatic filariasis: a manual for national elimination programmes. Geneva: World Health Organization. WHO/HTM/NTD/PCT/2011.4 WHO/HTM/NTD/PCT/ 2011.4.
- Sun D-J, Fang Y, Huang Y, Zhang Y (2020) Contributions to the lymphatic filariasis elimination programme and post-elimination surveillance in China by NIPD-CTDR. Adv Parasitol 110: 145–183. https://doi.org/10.1016/bs.apar.2020.04.007 PMID: 32563324
- 8. Rebollo MP, Bockarie MJ (2013) Rapid diagnostics for the endgame in lymphatic filariasis elimination. Am J Trop Med Hyg 89: 3–4. https://doi.org/10.4269/ajtmh.13-0202 PMID: 23843491
- Weil GJ, Ramzy RM (2007) Diagnostic tools for filariasis elimination programs. Trends Parasitol. https://doi.org/10.1016/j.pt.2006.12.001 PMID: 17174604
- Chu BK, Deming M, Biritwum NK, Bougma WR, Dorkenoo AM, et al. (2013) Transmission assessment surveys (TAS) to define endpoints for lymphatic filariasis mass drug administration: a multicenter evaluation. PLoS Negl Trop Dis 7: e2584. https://doi.org/10.1371/journal.pntd.0002584 PMID: 24340120
- Rao RU, Samarasekera SD, Nagodavithana KC, Punchihewa MW, Ranasinghe USB, et al. (2019) Systematic sampling of adults as a sensitive means of detecting persistence of lymphatic filariasis following mass drug administration in Sri Lanka. PLoS Negl Trop Dis 13: e0007365. <u>https://doi.org/10.1371/journal.pntd.0007365</u> PMID: 31009482
- Weil GJ, Curtis KC, Fakoli L, Fischer K, Gankpala L, et al. (2013) Laboratory and field evaluation of a new rapid test for detecting Wuchereria bancrofti antigen in human blood. Am J Trop Med Hyg 89: 11– 15. https://doi.org/10.4269/ajtmh.13-0089 PMID: 23690552
- Chesnais CB, Awaca-Uvon NP, Bolay FK, Boussinesq M, Fischer PU, et al. (2017) A multi-center field study of two point-of-care tests for circulating Wuchereria bancrofti antigenemia in Africa. PLoS Negl Trop Dis 11: e0005703. https://doi.org/10.1371/journal.pntd.0005703 PMID: 28892473
- Coffeng LE, Stolk WA, Golden A, de Los Santos T, Domingo GJ, et al. (2019) Predictive value of Ov16 antibody prevalence in different subpopulations for elimination of African onchocerciasis. Am J Epidemiol 188: 1723–1732. https://doi.org/10.1093/aje/kwz109 PMID: 31062838
- Prada JM, Davis EL, Touloupou P, Stolk WA, Kontoroupis P, et al. (2020) Elimination or Resurgence: Modelling Lymphatic Filariasis After Reaching the 1% Microfilaremia Prevalence Threshold. J Infect Dis 221: S503–S509. https://doi.org/10.1093/infdis/jiz647 PMID: 31853554
- Stolk WA, de Vlas SJ, Borsboom GJ, Habberna JD (2008) LYMFASIM, a simulation model for predicting the impact of lymphatic filariasis control: quantification for African villages. Parasitology 135: 1583– 1598. https://doi.org/10.1017/S0031182008000437 PMID: 19006602
- Subramanian S, Stolk WA, Ramaiah KD, Plaisier AP, Krishnamoorthy K, et al. (2004) The dynamics of *Wuchereria bancrofti* infection: a model-based analysis of longitudinal data from Pondicherry, India. Parasitology 128: 467–482. https://doi.org/10.1017/s0031182004004822 PMID: 15180315
- Eneanya OA, Gankpala L, Goss CW, Bolay FK, Weil GJ, et al. (2021) Impact of annual versus semiannual mass drug administration with ivermectin and albendazole on helminth infections in southeastern Liberia. Am J Trop Med Hyg 106. https://doi.org/10.4269/ajtmh.21-0768 PMID: 34814104
- Behrend MR, Basanez MG, Hamley JID, Porco TC, Stolk WA, et al. (2020) Modelling for policy: the five principles of the Neglected Tropical Diseases Modelling Consortium. PLoS Negl Trop Dis 14: e0008033. https://doi.org/10.1371/journal.pntd.0008033 PMID: 32271755
- **20.** Cao WC, Van der Ploeg CPB, Plaisier AP, van der Sluijs IJS, Habbema JDF (1997) Ivermectin for the chemotherapy of bancroftian filariasis: a meta-analysis of the effect of single treatment. Trop Med Int Health 2: 393–403. https://doi.org/10.1111/j.1365-3156.1997.tb00157.x PMID: 9171850
- Dreyer G, Addiss D, Noroes J, Amaral F, Rocha A, et al. (1996) Ultrasonographic assessment of the adulticidal efficacy of repeat high-dose ivermectin in bancroftian filariasis. Trop Med Int Health 1: 427– 432. https://doi.org/10.1046/j.1365-3156.1996.d01-79.x PMID: 8765448
- 22. Collyer BS, Irvine MA, Hollingsworth TD, Bradley M, Anderson RM (2020) Defining a prevalence level to describe the elimination of Lymphatic Filariasis (LF) transmission and designing monitoring & evaluating (M&E) programmes post the cessation of mass drug administration (MDA). PLoS Negl Trop Dis 14: e0008644.
- Irvine MA, Njenga SM, Gunawardena S, Njeri Wamae C, Cano J, et al. (2016) Understanding the relationship between prevalence of microfilariae and antigenaemia using a model of lymphatic filariasis infection. Trans R Soc Trop Med Hyg 110: 118–124. <u>https://doi.org/10.1093/trstmh/trv096</u> PMID: 26822604

- 24. Rao RU, Nagodavithana KC, Samarasekera SD, Wijegunawardana AD, Premakumara WD, et al. (2014) A comprehensive assessment of lymphatic filariasis inSri Lanka six years after cessation of mass drug administration. PLoS Negl Trop Dis 8: e3281. <u>https://doi.org/10.1371/journal.pntd.0003281</u> PMID: 25393404
- 25. Rao RU, Samarasekera SD, Nagodavithana KC, Punchihewa MW, Dassanayaka TD, et al. (2016) Programmatic Use of Molecular Xenomonitoring at the Level of Evaluation Units to Assess Persistence of Lymphatic Filariasis in Sri Lanka. PLoS Negl Trop Dis 10: e0004722. <u>https://doi.org/10.1371/journal.pntd.0004722</u> PMID: 27196431
- 26. Rao RU, Samarasekera SD, Nagodavithana KC, Goss CW, Punchihewa MW, et al. (2018) Comprehensive Assessment of a Hotspot with Persistent Bancroftian Filariasis in Coastal Sri Lanka. Am J Trop Med Hyg 99: 735–742. https://doi.org/10.4269/ajtmh.18-0169 PMID: 30014812
- Lau CL, Sheridan S, Ryan S, Roineau M, Andreosso A, et al. (2017) Detecting and confirming residual hotspots of lymphatic filariasis transmission in American Samoa 8 years after stopping mass drug administration. PLoS Negl Trop Dis 11: e0005914. <u>https://doi.org/10.1371/journal.pntd.0005914</u> PMID: 28922418
- Sheel M, Sheridan S, Gass K, Won K, Fuimaono S, et al. (2018) Identifying residual transmission of lymphatic filariasis after mass drug administration: Comparing school-based versus community-based surveillance—American Samoa, 2016. PLoS Negl Trop Dis 12: e0006583. <u>https://doi.org/10.1371/journal.pntd.0006583</u> PMID: 30011276
- Duerr HP, Raddatz G, Eichner M (2011) Control of onchocerciasis in Africa: threshold shifts, breakpoints and rules for elimination. Int J Parasitol 41: 581–589. <u>https://doi.org/10.1016/j.ijpara.2010.12</u>. 009 PMID: 21255577
- 30. de Vos AS, Stolk WA, de Vlas SJ, Coffeng LE (2018) The effect of assortative mixing on stability of low helminth transmission levels and on the impact of mass drug administration: Model explorations for onchocerciasis. PLoS Negl Trop Dis 12: e0006624. https://doi.org/10.1371/journal.pntd.0006624 PMID: 30296264